Author’s response to reviews

Title: A CASE OF TWENTY AND HALF SYNDROME: A CASE REPORT

Authors:

Mukesh Dube (dube_mukesh@yahoo.com)
Ronak Dani (ronak.dani@gmail.com)
Ayush Dubey (ayushdubey2@yahoo.co.in)
Dinesh Chouksey (dineshmd09@yahoo.com)

Version: 2 Date: 07 Jan 2019

Author’s response to reviews:

Respected Sir, I thank you for your mail. Two points have been raised regarding manuscript JMCR-D-18-00737R2.

The first point wants the title to show what type of study it is. The title A CASE OF TWENTY AND HALF SYNDROME: A CASE REPORT, already mentions it to be a case report.

Point 2 raises the query about author names and affiliations, which is already mentioned in all the author listings.

I hope you will kindly oblige. Thanking you, Dr. Mukesh Dube